Author's response to reviews

Title: Development of a validated patient reported symptom metric for pediatric Eosinophilic Esophagitis: qualitative methods

Authors:

James P Franciosi (james.franciosi@cchmc.org)
Kevin A Hommel (kevin.hommel@cchmc.org)
Charles W DeBrosse (charles.debrosse@cchmc.org)
Allison B Greenberg (allison.greenberg@cchmc.org)
Alexandria J Greenler (alexandria.greenler@cchmc.org)
J Pablo Abonia (pablo.abonia@cchmc.org)
Marc E Rothenberg (marc.rothenberg@cchmc.org)
Jame W Varni (JVarni@archmail.tamu.edu)

Version: 5 Date: 16 November 2011

Author’s response to reviews:

Dear Members of the BMC Gastroenterology Editorial Staff,

We have carefully reviewed the manuscript, and we have addressed major as well as minor formatting revisions. Please let us know if there is any additional information that we can provide. Thank you for all of your assistance, and we look forward to seeing our manuscript in BioMed Gastro.

Sincerely,

James P. Franciosi, MD, MS, MSCE, FAAP